Author's response to reviews

Title: Information needs and Seeking Behavior among health professionals working at governmental hospital and health centers in Bahir Dar, Ethiopia.

Authors:

Mulusew Asemahagn Andualem Mr (muler.hi@gmail.com)
Gashaw Kebede Dr (gashawkbd@yahoo.com)
Abera Kumie Dr (aberakumie2@yahoo.com)

Version: 6 Date: 14 December 2013

Author's response to reviews: see over
To BMC health service research journal editors

I have read my work again and again and I have some minor editorial corrections on the revised manuscript. These minor corrections or edits are:

1. On the title, it is to mean **governmental Hospital** not governmental hospitals
2. On background part, paragraph 1, line 4-5, it is was ….. health facilities and now corrected as … *in health facilities*.
3. On page 5, paragraph 3, line 1, the sentence construction is reconstructed as **Quantitative data were collected using pretested self-administered questionnaire**.
4. On page 6, last paragraph, line 1, it was “data form the……” and corrected as data from the……
5. On page 7 paragraph 2, line 2-3, it was “About 45% of the respondents was I the age group of……”, it is to mean About 45% of the respondents were in the age group of 26-30 years.